# University of Nebraska - Lincoln DigitalCommons@University of Nebraska - Lincoln

Publications from USDA-ARS / UNL Faculty

U.S. Department of Agriculture: Agricultural Research Service, Lincoln, Nebraska

3-1-2023

# The Impact of Crop Rotation and Spatially Varying Crop Parameters in the E3SM Land Model (ELMv2)

Eva Sinha Pacific Northwest National Laboratory

Ben Bond-Lamberty Pacific Northwest National Laboratory

Katherine V. Calvin Pacific Northwest National Laboratory

Beth A. Drewniak Argonne National Laboratory

Gautam Bisht Pacific Northwest National Laboratory

See next page for additional authors

Follow this and additional works at: https://digitalcommons.unl.edu/usdaarsfacpub

Part of the Agriculture Commons

Sinha, Eva; Bond-Lamberty, Ben; Calvin, Katherine V.; Drewniak, Beth A.; Bisht, Gautam; Bernacchi, Carl; Blakely, Bethany J.; and Moore, Caitlin E., "The Impact of Crop Rotation and Spatially Varying Crop Parameters in the E3SM Land Model (ELMv2)" (2023). *Publications from USDA-ARS / UNL Faculty*. 2602. https://digitalcommons.unl.edu/usdaarsfacpub/2602

This Article is brought to you for free and open access by the U.S. Department of Agriculture: Agricultural Research Service, Lincoln, Nebraska at DigitalCommons@University of Nebraska - Lincoln. It has been accepted for inclusion in Publications from USDA-ARS / UNL Faculty by an authorized administrator of DigitalCommons@University of Nebraska - Lincoln.

## Authors

Eva Sinha, Ben Bond-Lamberty, Katherine V. Calvin, Beth A. Drewniak, Gautam Bisht, Carl Bernacchi, Bethany J. Blakely, and Caitlin E. Moore



# **JGR** Biogeosciences

## **RESEARCH ARTICLE**

10.1029/2022JG007187

#### **Special Section:**

Advances in scaling and modeling of land-atmosphere interactions

#### **Key Points:**

- This study implements corn soybean rotation and spatially varying crop parameters in the Energy Exascale Earth System Land Model
- The model is calibrated and validated against observations collected at three AmeriFlux sites in the US Midwest
- We find that spatially varying crop parameters resulted in improved flux estimation from cropland areas

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

E. Sinha, eva.sinha@pnnl.gov

#### **Citation:**

Sinha, E., Bond-Lamberty, B., Calvin, K. V., Drewniak, B. A., Bisht, G., Bernacchi, C., et al. (2023). The impact of crop rotation and spatially varying crop parameters in the E3SM land model (ELMv2). Journal of Geophysical Research: Biogeosciences, 128, e2022JG007187. https://doi. org/10.1029/2022JG007187

Received 14 SEP 2022 Accepted 2 MAR 2023

#### **Author Contributions:**

**Conceptualization:** Eva Sinha, Ben Bond-Lamberty, Katherine V. Calvin **Formal analysis:** Eva Sinha

© 2023 UChicago Argonne, LLC, Battelle Memorial Institute and The Authors. This article has been contributed to by U.S. Government employees and their work is in the public domain in the USA.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# The Impact of Crop Rotation and Spatially Varying Crop Parameters in the E3SM Land Model (ELMv2)

Eva Sinha<sup>1</sup>, Ben Bond-Lamberty<sup>2</sup>, Katherine V. Calvin<sup>2</sup>, Beth A. Drewniak<sup>3</sup>, Gautam Bisht<sup>1</sup>, Carl Bernacchi<sup>4,5</sup>, Bethany J. Blakely<sup>5</sup>, and Caitlin E. Moore<sup>5,6</sup>

<sup>1</sup>Pacific Northwest National Laboratory, Richland, WA, USA, <sup>2</sup>Pacific Northwest National Laboratory, Joint Global Change Research Institute, College Park, MD, USA, <sup>3</sup>Argonne National Laboratory, Lemont, IL, USA, <sup>4</sup>Global Change and Photosynthesis Research Unit, USDA-ARS, Urbana, IL, USA, <sup>5</sup>University of Illinois at Urbana-Champaign, Urbana, IL, USA, <sup>6</sup>School of Agriculture and Environment, The University of Western Australia, Crawley, WA, Australia

**Abstract** Earth System Models (ESMs) are increasingly representing agriculture due to its impact on biogeochemical cycles, local and regional climate, and fundamental importance for human society. Realistic large scale simulations may require spatially varying crop parameters that capture crop growth at various scales and among different cultivars, as well as common crop management practices, but their importance is uncertain, and they are often not represented in ESMs. In this study, we examine the impact of using constant versus spatially varying crop parameters using a novel, realistic crop rotation scenario in the Energy Exascale Earth System Model (E3SM) Land Model version 2 (ELMv2). We implemented crop rotation by using ELMv2's dynamic land unit capability, and then calibrated and validated the model against observations collected at three AmeriFlux sites in the US Midwest with corn soybean rotation. The calibrated model closely captured the magnitude and observed seasonality of carbon and energy fluxes across crops and sites. We performed regional simulations for the US Midwest using the calibrated model and found that spatially varying only a few crop parameters across the region, as opposed to using constant parameters, had a large impact, with the carbon fluxes and energy fluxes both varying by up to 40%. These results imply that large scale ESM simulations using spatially invariant crop parameters may result in biased energy and carbon fluxes estimation from agricultural land, and underline the importance of improving human-earth systems interactions in ESMs.

**Plain Language Summary** Crops are increasingly being characterized in global land models because of their impact on local and regional climate. However, there is limited understanding of the impact of crop rotation and of different crop cultivars on carbon and energy fluxes from the land surface. Our study implements crop rotation and spatially varying crop parameters in the Energy Exascale Earth System Model Land Model and finds that doing so improves carbon and energy flux estimation from cropland area. These findings emphasize the importance of capturing agricultural management practices and variability in growth characteristics across different crop cultivars in global land models.

#### 1. Introduction

Agriculture affects local, regional, and global climate through greenhouse gas emissions, and modifications of the biogeochemical cycles, water and energy budget (McDermid et al., 2017). Agricultural land conversion, expansion, and intensification is a major source of greenhouse gas (GHG) emissions, contributing 23% of the total anthropogenic emissions of GHGs (IPCC, 2019). Agricultural intensification also impacts local and regional temperature and precipitation via modification of surface energy partitioning and an increase in evapotranspiration (Lobell et al., 2006; Mueller et al., 2016; D. Lombardozzi et al., 2018). Due to these impacts, and because of its fundamental importance to human societies and well-being, agriculture is increasingly being represented in Earth System Models (ESMs) (Drewniak et al., 2013; Levis et al., 2012; Liu et al., 2016; Osborne et al., 2015; Wu et al., 2016).

ESMs however, lack adequate representation of crop rotation, a management practice that is dominant in North America and common worldwide (Sahajpal et al., 2014; Wallander, 2013). Crop rotation, where different crops are grown on the same land across a sequence of growing seasons, has been practiced since historical times as it improves soil quality (Karlen et al., 2006), increases soil carbon sequestration (West & Post, 2002), enhances microbial richness and diversity (Venter et al., 2016), and increases crop yield while reducing fertilizer





Funding acquisition: Ben Bond-Lamberty, Katherine V. Calvin Investigation: Eva Sinha Methodology: Eva Sinha, Ben Bond-Lamberty, Katherine V. Calvin, Gautam Bisht

Project Administration: Ben Bond-Lamberty, Katherine V. Calvin Resources: Eva Sinha, Beth A. Drewniak, Gautam Bisht, Carl Bernacchi, Bethany J. Blakely, Caitlin E. Moore Supervision: Ben Bond-Lamberty, Katherine V. Calvin, Beth A. Drewniak Validation: Eva Sinha Visualization: Eva Sinha Writing – original draft: Eva Sinha Writing – review & editing: Eva Sinha, Ben Bond-Lamberty, Beth A. Drewniak, Gautam Bisht, Carl Bernacchi, Caitlin E. Moore requirements (Bowles et al., 2020; Smith et al., 2008; Stanger & Lauer, 2008). Crop rotation can also help mitigate and adapt to climate change due to its potential for carbon sequestration and reducing nitrogen loss from agricultural systems (Lal et al., 2011; Wang et al., 2010).

Crop rotation is increasingly being implemented in land model components of ESMs. Sequential cropping, where multiple crops are grown on the same land in a given year, has been implemented in CLM5.0 for site level data in central Europe (Boas et al., 2021) and in Joint UK Land Environment Simulator (JULES) at site-level in France and regional level in India (Mathison et al., 2021). Crop rotation was also previously implemented in CLM5.0 for a single site in the US Midwest (Cheng et al., 2020). All of these studies modified the crop parameters based on values in the literature, field observation, or calibration using a simple one-at-time approach that varies a single model parameter at a time. This simplistic parameterization approach, however, fails to account for the impact of joint parameter variability on model outputs (Qian et al., 2018; Ricciuto et al., 2018) and may thus fail to accurately capture fluxes from croplands.

Adequate crop representation depends on calibrating various crop parameters, and large scale ESM simulations likely require parameters that are scale- and cultivar-dependent. Calibrating ESM crop parameters using site-level observations is challenging due to the limited availability of observational data and the computational cost involved with model calibration and validation. Due to these limitations static or spatially invariant crop parameters are often used for regional/global runs (Osborne et al., 2009; Levis et al., 2012; Drewniak et al., 2013; D. L. Lombardozzi et al., 2020). However, crop parameters can be scale- and cultivar-dependent (Iizumi et al., 2014; Mohammadi, 2007) and therefore spatially invariant parameters can result in biases between observed and simulated fluxes and are therefore not recommended for large scale simulations (Iizumi et al., 2014). The prevalence and magnitude of such a bias are poorly understood, however.

The objective of this study is to understand the impact of using constant versus varying crop parameters in a realistic crop-rotation scenario, and quantify the resulting model's fidelity against high-quality AmeriFlux observational data. To do so, we enhance the crop modeling capability of the Energy Exascale Earth System Model (E3SM) land component version 2 (ELMv2) by implementing corn soybean rotation in ELM; calibrate and validate the model using observations from multiple sites; and quantify the impact of different parameterization schemes on carbon and energy fluxes in a regional North America simulation.

## 2. Methodology

#### 2.1. ELM Crop Model

The E3SM land model version 2 (ELMv2) is branched from CLM version 4.5 (CLM4.5) (Oleson et al., 2013). The major additions to ELM since diverging from CLM4.5 are described in detail in Golaz et al. (2022), Burrows et al. (2020), and Ricciuto et al. (2018); they include improved representation of atmospheric aerosols, a minor bug fix in evaporation estimation from pervious surfaces, an updated scheme for calculation of leaf stomatal conductance, and modification to the nighttime albedo calculation. The ELM crop model includes representation of major crop types in order to capture the biogeochemical and biophysical impact of crops on land surfaces (Drewniak et al., 2013; Levis et al., 2012). To date it has not included any crop rotation capability.

#### 2.2. Corn Soybean Rotation Implementation

We implemented a corn soybean rotation, the most common such rotation type in North America (Wallander, 2013), in ELM by using the model's *dynamic land unit* capability. Similar to the Community Land Model (CLM) version 5.0 (Lawrence et al., 2019), dynamic land units allow for the fraction of crop functional types (cfts) in each soil column to be adjusted over time, as specified in the model's input land use time series. Modifying the cfts percentage from 1 year to another thus results in a realistic rotation between two or more crops. For instance, corn soybean rotation for site-scale simulation was represented in the land use time series by switching from 100% corn to 100% soybean for the crop rotation years. A corn soybean rotation for our regional simulation was implemented in the land use time series based on the Land-Use Harmonization 2 (LUH2) transition data set and is described in Section 2.4.1.

**Ohio Valley** 

21698961, 2023, 3, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2022JG007187, Wiley Online Library on [26/04/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/term



100

Northern Rockies

2.3. Site-Scale Calibration and Validation

0



Upper Midwest

Figure 1. Location of AmeriFlux observational sites and three sub-regions of the US Midwest used for the regional run. Observational sites used for site level calibration and validation are shown in red and sites used for regional validation are

The model was calibrated and validated based on observations from three corn soybean rotation sites in the US Midwest (Figure 1). Site location and mean climatic conditions are summarized in Table 1; all sites are rainfed, that is, have no irrigation. At the US-Ne3 and US-Ro1 sites, rotation occurred every year between 2001-2014 and 2004–2016, respectively. At the US-UiC site, the rotation consisted of two years of corn plantation followed by one year of soybean plantation, from 2008 to 2016. Meteorological forcing data collected at the three sites including, air temperature, precipitation, downwelling shortwave radiation, downwelling longwave radiation, humidity,

Observational Sites Used for Site Level Calibration and Validation and Regional Validation

shown in green.

2.3.1. Site Level Data



Table 1

conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

air pressure, and wind speed, was utilized for model simulation. For US-Ne3 we used meteorological forcing data collected between 2002 and 2015, for US-Ro1 between 2009 and 2012, and for US-UiC between 2011 and 2016.

The data for the US-Ne3 site is part of the FLUXNET 2015 data set that was gap-filled and processed based on the methodology described in Pastorello et al. (2020). The gap filled and partitioned data for the US-Ro1 site was downloaded from the AmeriFlux website (downloaded in April 2021). For the US-UiC site, the gross primary productivity (GPP) and ecosystem respiration (ER) were calculated using standard methodologies from net ecosystem exchange values measured at the eddy covariance flux towers (Moore et al., 2020). The flux tower derived GPP and ER are referred to as observed GPP and ER, respectively, in the remainder of the manuscript. The US-UiC data is not currently available on the AmeriFlux website, but will be in the future. We converted the half-hourly and hourly data from these sites into daily averages for calibrating and validating ELMv2.

#### 2.3.2. Model Calibration

We used carbon and energy flux measurements at the three sites to calibrate the model, and leaf area index (LAI), canopy height, and harvest yield to validate it. Simulated harvest yield here refers to grain harvest and captures the carbon flux into the grain pool.

The model parameters were calibrated similar to Sinha et al. (2022) by first developing an ELM surrogate model across a range of input parameters, followed by sensitivity analysis to identify the most influential parameters, and lastly performing Bayesian calibration of these surrogate models to find optimum ELM parameter values.

We identified 12 crop parameters related to crop phenology, crop management, CN allocation, and photosynthetic capacity whose parameters values are most uncertain. The input range for these parameters was identified based on literature review and expert judgment (Table 4). The parameter values for these parameters were randomly varied over their uniform prior range to generate 2,000 ELM simulations. The default value for other crop parameters that were not optimized are listed in Table S1 in Supporting Information S1. ELM simulations were submitted via the Offline Land Model Testbed (Ricciuto, 2022) and each ran for 200 years in the accelerated spin-up mode, 200 years in the non-accelerated spin-up mode (Thornton & Rosenbloom, 2005), and 165 years in transient mode from 1850 to 2015. The 2,000 spin-up simulations were performed to bring the carbon pool to equilibrium for 1850 climatic conditions. Crops were then activated in the transient simulations by using the land use time series containing information on land cover change and crop rotation. Corn and soybean crops were rotated for eight to fourteen years corresponding to the observed crop rotation years at the AmeriFlux sites. At all three AmeriFlux sites, c3 grass was simulated for years prior to the start of crop rotation. For the spin-up and transient simulations, the available forcing data for each site was recycled. The model output was postprocessed for four output Quantity of Interests (QoIs)-gross primary productivity (GPP), ecosystem respiration (ER), latent heat flux (LE), and sensible heat flux (H). The post processing involved estimating daily average over the last 10 years of the transient run for the four QoIs that were then used for developing surrogates for each day of the year for the four QoIs for each crop and for each site. Similar to Sinha et al. (2022)'s approach, 1600 ELM simulations were used for developing the surrogates and 400 were used for testing the accuracy of surrogates.

Sobol sensitivity indices were used to examine parametric uncertainty (Saltelli et al., 2010; Sobol, 2001) and identify the most influential parameters for reducing this uncertainty in model outputs. We evaluated the main effect sensitivity that estimates the contribution of one parameter at a time to the total variance in the output variable.

Model parameters were calibrated to better match the model outputs to observations. We used Markov chain Monte Carlo (MCMC) to sample the parameter input space and reduce the bias between model output and observations. MCMC's requirement of large number of model evaluations was met by using computationally inexpensive surrogate models (see above) instead of computationally expensive ELM model outputs. Calibration was performed simultaneously for all four QoIs to identify a single set of parameter values for each crop. We limited the maximum number of parameters for calibration to five, and selected parameters for which the probability density function of the optimized parameter was normally distributed within the input range instead of being skewed to either side of the input range. Since GPP is low or negligible during the non-growth period, we calibrated it only during the growth period; this GPP calibration window was from May 8 to October 10 for US-Ne3 corn; May 29 to September 26 for US-Ne3 soybean; June 3 to October 2 for US-Ro1 corn; June 19 to September 12 for US-Ro1 soybean; April 30 to October 27 for US-UiC corn; and April 30 to October 27 for US-UiC soybean. The other three QoIs were calibrated using observations for all days in the year.



Table 2

Calibration and Validation Years

		Co	orn	Soybean			
Usage	Site	Calibration years	Validation years	Calibration years	Validation years		
Site level calibration/ validation	US-Ne3	2001, 2007, 2009, 2011, 2013	2003, 2005	2002, 2008, 2010, 2012, 2014	2004, 2006		
	US-Ro1	2005, 2007, 2009, 2011	2013, 2015	2004, 2006, 2008, 2010, 2012	2014, 2016		
	US-UiC	2009, 2011, 2012	2014, 2015	2010, 2013	2016		
Regional validation	US-Bo1	-	1997, 1999, 2001, 2003, 2005, 2007	-	1998, 2000, 2002, 2004, 2006, 2008		
	US-Br1	-	2005, 2007, 2009, 2011	-	2006, 2008, 2010		
	US-IB1	-	2006, 2008, 2010, 2012, 2014, 2016, 2018	-	2005, 2007, 2009, 2011, 2013, 2015, 2017		

#### 2.3.3. Model Validation

For each site-crop, 2 years of carbon flux, energy flux, leaf area index (LAI), canopy height, and harvest measurements were used for model validation (Table 2).

The optimized parameter obtained from model calibration were utilized for running a single model simulation for each site. Similar to the calibration runs, the validation simulation ran for 200 years in the accelerated spin-up mode, 200 years in the nonaccelerated spin-up mode, followed by a transient run from 1850 to 2015 that used site specific meteorological data. For performing model validation, simulated carbon fluxes and energy fluxes were compared to the observations for the validation years, while LAI and annual harvest yield were compared to observations for all years since LAI and harvest were not utilized for model calibration.

#### 2.4. Regional Analysis

#### 2.4.1. Generation of Corn Soybean Rotation Historical Landuse Timeseries

Corn soybean rotation was represented in the historical land use time series from 2000—2015 based on information in the LUH2 historical transition data set (Hurtt et al., 2020). LUH2 provides landuse transition information at an annual temporal resolution and at  $0.25^{\circ}$  spatial resolution between five cfts: C3 annuals, C4 annuals, C3 perennials, C4 perennials, and C3 nitrogen fixers. Several crop types are aggregated into each of these five cfts. For the US-Midwest, the crop with the largest harvested acres in the C4 annual cft is corn while in the C3 nitrogen fixer cft is soybean. Therefore, the LUH2 historical transition from C4 annual (*c4ann*) to C3 nitrogen fixer (*c3nfx*) was utilized for generating corn soybean rotation for the US-Midwest between 2000 and 2015. This transition is represented as unit fraction per gridcell (*frac*<sub>c4ann\_to\_c3nfx</sub>). The corn soybean rotation was implemented in the landuse timeseries by, first, identifying grid cells with *frac*<sub>c4ann\_to\_c3nfx</sub> greater than 5% within the United States (Figure S1c in Supporting Information S1). Second, the fraction of corn or soybean in ELM land use timeseries gridcell was modified such that in even years between 2000 and 2015 the fraction of soybean was transferred to corn, while in odd years the fraction of corn was transferred to soybean (Equations 1 and 3). During this time period, the fractions of soybean in even years and corn in odd years were reduced corresponding to the increase in the other crop; this maintained the total corn and soybean area (Equations 2 and 4). Prior to 2000, crop rotation was not implemented in the landuse time series.

For even years between 2000 and 2015:

$$Corn = Corn + frac_{c4ann\_to\_c3nfx} \times Soybean$$
(1)

$$Soybean = (1 - frac_{c4ann\_to\_c3nfx}) \times Soybean$$

For odd years between 2000 and 2015:

1

$$Soybean = Soybean + frac_{c4ann\_to\_c3nfx} \times Corn$$

$$Corn = (1 - frac_{c4ann\_to\_c3nfx}) \times Corn$$

(2)

(3)

(4)



Table 3         Crop Parameters Used for Different Sets and Regions							
Set/Sub-regions	Northern Rockies	Upper Midwest	Ohio Valley				
Set1	for all sub-regions	based on US-Ne3 optimization	n				
Set2	for all sub-regions based on US-Ro1 optimization						
Set3	for all sub-regions	based on US-UiC optimization	n				
Composite	US-Ne3 optimization	US-Ro1 optimization	US-UiC optimization				

#### 2.4.2. Regional Simulation

Regional simulations were performed for the Corn Belt in the US Midwest divided into three sub-regions: Northern Rockies, Upper Midwest, and Ohio Valley. These sub-regions were roughly based on the NOAA's US climatic regions (https://www.ncdc.noaa.gov/monitoring-references/maps/us-climate-regions) and each sub-region contained one of the calibration sites (Figure 1). Importantly, we used these sub-regions only to demonstrate the impact of constant versus spatially varying parameters; they do not represent fixed application boundaries for the optimized parameters. We performed five regional simulations using the optimized parameters obtained from the three calibration sites. Of these, three (Set1, Set2, and Set3) used the same crop parameters for all three regions, while the fourth set (Composite) incorporated varying crop parameters from all three sub-regions (Table 3). Finally, a fifth regional simulation (Default) was performed with default (uncalibrated) ELM crop parameters. Corn soybean rotation was implemented in all five regional simulations.

Similar to the site-level simulations, the regional simulations involved running the model in accelerated spin-up mode for 200 years, followed by nonaccelerated spin-up mode for 200 years, and transient run from 1850 to 2015. The spin-up simulations were performed to bring the carbon pool to equilibrium for 1850 climatic conditions; crops were then activated in the transient simulations by using the land use time series containing information on land cover change and crop rotation. For the spin-up simulations, natural vegetation consisting of c3 grass,

Table 4

Descriptions, Input Ranges, and Sources of Information Used for the Twelve Input Parameters Varied in This Study

				De	Default Range			
Parameter	ELM variable	Units	Description	Corn	Soybean	Corn	Soybean	Source
$\overline{T}_p$	planting_temp	К	Average 10-day temperature required for plant emergence	287	288	287–293	287–293	1
$\overline{d}_L$	declfact	-	Decline factor for gddmaturity	1.05	1.05	0.7-1.575	0.7-1.575	1
	fertnitro	kgN m <sup>-2</sup>	Maximum fertilizer to be applied	0.015	0.0025	0.01-0.02	0.002-0.003	1
Ī <sub>emerg</sub>	lfemerg	_	Leaf emergence parameter	0.03	0.03	0.01-0.05	0.01-0.05	1
	mxmat	_	Maximum number of days to maturity	165	150	125-175	125-175	1
$\overline{GDD}_{hyb}$	hybgdd	°day	Growing degree days required for maturity	1,700	1,900	1,275–2,125	1,425–2,375	2
	leafcn	gC gN <sup>-1</sup>	Leaf CN ratio	25	25	8-25	8-25	3
$\overline{L}_{\max}$	laimx	-	Maximum leaf area index	5	6	4–7	3–7	4
SLA	slatop	$m^2 gC^{-1}$	Specific leaf area (SLA) at top of canopy, projected area basis	0.05	0.07	0.03-0.08	0.02-0.07	5
	$br_mr \times 10^{-6}$	umol $CO_2 m^{-2} s^{-1}$	Base rate for maintenance respiration (MR)	2.52	2.52	1.26-3.75	1.26-3.75	6
	q10_mr	-	Temperature sensitivity for MR	2.2	2.2	1.3-3.3	1.3-3.3	6
	mbbopt	-	Ball-Berry model equation slope	4.0	9.0	4–12	4–12	7

*Note.* The ranges are based on (a) expert judgment (in the case where there is insufficient literature, but within 25% of the default value would be inappropriate) (b) within 25% of the default value; (c) Srivastava et al. (2006) and Li et al. (2019) (d) Baez-Gonzalez et al. (2005) and Nguy-Robertson et al. (2012) (e) Nagasuga et al. (2014) (f) Ricciuto et al. (2018) (g) Personal communication with Dr. Dan Ricciuto.



Table 5									
Most Sensitive Parameters and Their Optimum Value After Calibration									
	Input range		US-Ne3		US-Ro1		US-UiC		
Parameter	Corn	Soybean	Corn	Soybean	Corn	Soybean	Corn	Soybean	
$br_mr \times 10^{-6}$	1.26–3.75	1.26–3.75	_	3.74	3.54	3.14	-	_	
hybgdd	1,275–2,125	1,425–2,375	1,383	1,428	-	1,434	1,331	1,425	
leafcn	8-25	8–25	18	22	25	22	25	25	
mbbopt	4–12	4–12	4.9	6.0	12.0	4.5	4.8	6.7	
mxmat	125-175	125-175	-	-	-	-	-	146	
planting_temp	287-293	287–293	290	291	291	290	291.7	289	
slatop	0.03-0.08	0.02-0.07	0.03	_	0.08	_	0.03	_	

c4 grass, and broadleaf deciduous temperate trees was used in place of croplands. For the regional simulations, meteorological forcing was based on the Global Soil Wetness Project Phase 3 (GSWP3) data. The GSWP3 data set was chosen since it has shown better agreement with benchmark for both forcing variables and CLM5 output variables, when forced with the GSWP3 forcing data set, compared to other forcing datasets (Lawrence et al., 2019). The GSWP3 forcing data is available from 1901 to 2014 and therefore for transient runs from 1850 to 1900 and spin-up simulations, GSWP3 forcing data from 1901 to 1920 was recycled the 1920–2014 transient runs utilized the GSWP3 data from that period.

ELM currently only accepts spatially and temporally constant crop parameters. Therefore, ELM outputs for the composite set with spatially varying crop parameters were generated by combining the outputs from Set1, Set2, and Set3; output value for grid cells within the Northern Rockies were obtained from Set1, for grid cells within the Upper Midwest from Set2, and for grid cells within the Ohio Valley from Set3.

#### 2.4.3. Regional Validation

Tabla 5

We used FluxCom (Jung et al., 2020) based GPP measurements for validating the GPP simulated for the US-Midwest. We compared model simulated GPP to a median of a 30 member ensemble of FluxCom GPP estimates based on remote sensing and meteorological data. Additionally, we validated regionally simulated GPP and LE to site level observations. We validated simulated GPP by comparing against the three AmeriFlux sites used for model calibration and validation (Section 2.3.1) and validated LE by comparing against these three sites and three additional AmeriFlux sites (US-Bo1, US-Br1, and US-IB1) in the US Midwest with corn soybean rotation (Figure 1 and Table 1). Similar to the US-Ro1 site, the gap filled and partitioned data for these three additional sites was downloaded from the AmeriFlux website (downloaded in April 2021). At the US-Bo1, US-Br1, and US-IB1 sites, corn-soybean rotation occurred every year between 1997 and 2008, 2005 and 2011, and 2005 and 2018, respectively (Table 2). Since, observations for the six AmeriFlux sites were available for different years hence, we validated regionally simulated GPP and LE by comparing observations across all years to transient simulations from 2001 to 2010.

## 3. Results

#### 3.1. Site-Scale Calibration and Validation

The most influential parameters for both corn and soybean were similar across the three sites, with three parameters (leafcn, mbbopt, and planting\_temp) being common across both crops and sites (Figures S2, S3, S4, S5 in Supporting Information S1 and Table 5). For both crops, the parameters controlling plant phenology (planting\_temp and hybgdd) were among the most influential parameters across all four QoIs, except hybgdd for US-Ro1. Another phenological parameter that controls the maximum number of days required to reach maturity, mxmat, was among the five most influential parameters for few sites, crops, and QoIs. Across all three sites, the parameter associated with stomatal conductance (mbbopt) was more sensitive for soybean than for corn. The parameters controlling leaf CN allocation (leafcn) and top of canopy specific leaf area (slatop) were both identified as influential parameters across crops and sites, with leafcn being generally more influential for carbon fluxes and slatop more influential for energy fluxes. The parameter controlling the base rate for



**T**. L L . 4

Table o	detter and Deet						
Parameters Usea for vali	US-Ne3		US	-Ro1	US-UiC		
Parameter	Corn	Soybean	Corn	Soybean	Corn	Soybean	
$br_mr \times 10^{-6}$	3	5.1 <sup>a</sup>	3.3ª		2.52 <sup>b</sup>		
hybgdd	1,400	1,400	1,700 <sup>b</sup>	1,400	1,300	1,400	
leafcn	18	22	25	22	25	25	
mbbopt	5	6	12	4.5	5	7	
mxmat	165 <sup>b</sup>	150 <sup>b</sup>	165 <sup>b</sup>	150 <sup>b</sup>	165 <sup>b</sup>	150	
planting_temp	290	291	291	290	291	289	
slatop	0.03	0.07 <sup>b</sup>	0.08	0.07 <sup>b</sup>	0.03	0.07 <sup>b</sup>	

<sup>a</sup>br\_mr parameter is not cft specific hence average of the optimized value for corn and soybean was used. <sup>b</sup>For parameters that were not optimized during calibration their default values were used.

maintenance respiration (br\_mr) was identified as the most influential parameter for ER across sites and crops during the non-growing season and less influential during the growing season. Since maintenance respiration is negligible during the non-growing season, br\_mr was selected among the most influential parameters for only few sites and crops (Table 5).

The optimized parameter values varied across sites and crops (Table 5) with parameters being more similar across the US-Ne3 and US-UiC site than the US-Ro1 site. The final parameter values were based on rounding off the optimized parameter value, averaging the optimized values for non-cft specific parameter (br\_mr), and using default values for parameters that were not optimized (Table 6). The US-Ro1 corn calibration resulted in optimized parameter values for mbbopt and slatop much higher than other crops and sites. For this site, meteorological forcing data was available for only few years that may have contributed to the higher estimated parameter values.

In general, the calibrated model captured the observed seasonality and magnitude of GPP, ER, and LE with the fraction of the variance explained being higher for carbon fluxes than energy fluxes. The calibrated GPP matched the observed seasonality and peak magnitude for both crops at the three sites except the timing of leaf senescence for both crops at US-Ro1 and the peak GPP for corn at US-UiC (subplots A and B in Figure 2, Figures S6 and S7 in Supporting Information S1). Overall, within the calibration window, the posterior GPP estimates explained 97% and 87% of the observed daily variance at the US-Ne3 site, 85% and 77% of variance at US-Ro1 site, and 93% and 89% of the variance at the US-UiC site for corn and soybean, respectively (Table S2 in Supporting Information S1). Across crops and sites, the calibrated ER matched the observed seasonality and the magnitude during the growing period, however the simulated magnitude differed from observations for most crop/sites during the non-growth period and for US-Ne3 soybean during the growing period (subplots C and D in Figure 2, Figures S6 and S7 in Supporting Information S1). The posterior ER estimates explained more than 80% of the observed daily variance across crops and sites (Table S2 in Supporting Information S1). The posterior estimates of latent heat flux captured the observed seasonality and magnitude, although sensible heat flux was not well calibrated, especially for soybean (subplots E—H in Figure 2, Figures S6 and S7 in Supporting Information S1).

The model closely captured the seasonality and peak magnitude of various fluxes across most, but not all, crops and sites (Figure 3, Figures S8 and S9 in Supporting Information S1). Seasonality of carbon and energy fluxes was well reproduced across crops and sites, except for the sensible heat flux for soybean and leaf senescence timing at US-Ro1. The peak flux magnitude for GPP, ER, and LE, was well captured for corn at all three sites, except ER at US-UiC; while for soybean the peak magnitude of these fluxes was underestimated. The markedly higher observed corn ER at US-UiC during the validation years, 51% and 57% higher than 10-year average (Moore et al., 2022), resulted in the large difference between simulation and observations (Figure S9c in Supporting Information S1). The model was unable to reproduce the peak magnitude for sensible heat flux across crops and sites, except for corn at US-UiC.

The simulated values of LAI and yield, outputs not used for calibration, were lower than observations. The simulated peak LAI magnitude was lower than observations across crops and sites, with the difference between





**Figure 2.** Model calibration: Observed versus prior and posterior distribution of the modeled GPP (gC  $m^{-2} day^{-1}$ ), ER (gC  $m^{-2} day^{-1}$ ), LE (W  $m^{-2}$ ), and, H (W  $m^{-2}$ ) for corn and soybean at US-Ne3. The prior distribution (red shade) represents the daily simulated values for the 2000 ensemble members while the posterior distribution (green shade) represents the calibrated values estimated with the optimized parameters. The black line represents observed average daily across the calibration years (Table 2).





**Figure 3.** Model validation: Observed versus simulated GPP (gC m<sup>-2</sup> day<sup>-1</sup>), ER (gC m<sup>-2</sup> day<sup>-1</sup>), LE (W m<sup>-2</sup>), and, H (W m<sup>-2</sup>) for corn and soybean at US-Ne3 using optimized parameter values (Table 6). The red lines represent daily average model simulation over the last 10 years of transient run and the black line represents observed daily average values over the validation years (Table 2). Relative Root Mean Square Error (RRMSE) is dimensionless and represents the root mean square error (RMSE) normalized by the root mean square observations.



Journal of Geophysical Research: Biogeosciences



**Figure 4.** Model validation: Observed versus simulated leaf area index (LAI) for corn and soybean at US-Ne3 using optimized parameter value (Table 6). The black lines represent simulated LAI over the calibration and validation years. The blue (corn) and orange (soybean) circles represents observed weekly LAI.

observed and simulated much smaller at the US-Ne3 site than at the other two sites (Figure 4, Figures S10 and S11 in Supporting Information S1). The simulated harvest captured the yearly harvest variability for the US-Ro1 site, was toward the lower end of the observed harvest for the US-UiC site, and was underestimated for the US-Ne3 site (Figure 5, Figures S12 and S13 in Supporting Information S1).

Simulations closely capture distinctly different GPP patterns between corn and soybean after implementation of crop rotation in ELM. At the US-Ne3 site, observed annual GPP for the soybean years was approximately 60% of the annual GPP for corn and this large variability between the two crops was accurately captured by ELM (Figure S14 in Supporting Information S1). At the US-Ne3 site 100% of crop is rotated whereas crop rotation occurs in less than 20% of the gridcell fraction in the regional simulation (Figure S1 in Supporting Information S1). In the regional simulation, at the cft level, annual GPP varies by approximately 20% in grid cells with maximum corn soybean rotation, while at the grid level the difference in annual GPP was negligible. This is because each



Figure 5. Model validation: Observed versus simulated crop harvest for corn and soybean using optimized parameter value (Table 6). The orange bars represent simulated annual harvest over the calibration and validation years and the gray bars represents observed harvest. Light yellow background represents corn years and light blue background represents soybean years.



# Journal of Geophysical Research: Biogeosciences

#### 10.1029/2022JG007187



**Figure 6.** Impact of constant versus varying parameters on annual GPP: Total annual gross primary productivity (GPP) estimated by using regionally varying parameters (a) difference in GPP (b–d) and percent difference in GPP (e–g) when using regional versus constant parameters for corn and soybean. Set1 is based on parameters obtained from calibrating US-Ne3, Set2 is based on US-Ro1 calibration, and Set3 is based US-UiC calibration. Composite set utilized parameters based on US-Ne3 calibration for the Northern Rockies, based on US-Ro1 calibration for the Upper Midwest, and based on US-UiC for the Ohio Valley (Figure 1). Annual simulated GPP are based on average of 10 years of transient runs from 2001 to 2010.

grid cell is comprised of several landunits that in turn consists of various cfts. In addition, only a fraction of the corn cft undergoes crop rotation. Thus, when crop rotation impact is scaled up to the gridcell level it becomes negligible.

#### 3.2. Regional Simulation

We found that varying only few parameters across the region had a large impact on carbon and energy fluxes. Annual GPP and ER varied by up to 40% and 35% (Figure 6 and Figure S16 in Supporting Information S1), respectively; the difference in fluxes was driven by both corn and soybean (Figure S15 in Supporting Information S1). Using non-regional parameters produced large changes in fluxes, both positive and negative. For example, grid cells in the Ohio Valley had both higher and lower fluxes in various regions when optimized crop parameters from the two other regions were utilized (subplots b, c, e, and f in Figure 6 and Figure S16 in Supporting Information S1).

Analogous to carbon fluxes, energy fluxes also varied across the regions due to difference in crop parameters. For the summer months from June to September, different crop parameters resulted in LE varying by up to 15% (Figure S17 in Supporting Information S1) and H varying by up to 40% (Figure S18 in Supporting Information S1). Additionally, the impact of different crop parameters varied across different months. For example, grid cells located in the Upper Midwest observe a lower LE flux in July and higher in August when optimized crop parameters from the Northern Rockies were used for their simulation (subplots f and j in Figure S17 in Supporting Information S1).

Optimized and spatially varying crop parameters reduced the difference between observed and simulated annual GPP as compared to the default uncalibrated parameters. At a regional scale, the absolute average difference in simulated and FluxCom (Jung et al., 2020) based annual GPP from 2001 to 2010 reduced from 46% to 25% when default crop parameters were replaced with calibrated and spatially-varying crop parameters (Figure 7). Site level comparison across the three calibration sites yielded similar results with annual GPP for corn(soybean) varying by an average of 43%(71%) compared to 1%(8%), for the default and calibrated and crop parameters, respectively (Figure 8). For the site level comparison, grid cells containing the observational sites were selected from the regional simulation. Similar to annual GPP, the relative root mean square error (RMSE) between simulated





Figure 7. Comparison of simulated annual GPP to FluxCom estimates: Annual GPP estimates based on FluxCom (Jung et al., 2020) (a), ELM default crop parameters (b), composite set with calibrated and spatially varying parameters (c), percent difference between FluxCom and default set (d), and percent difference between FluxCom and composite set (e). FluxCom GPP are based on average over 2001–2010 and simulated GPP are based on average of 10 years of transient runs from 2001 to 2010.

and observed LE was lower when calibrated crop parameters were used as compared to default for 7 out of 12 crop-sites (Figure S20 in Supporting Information S1). Site level comparison of monthly fluxes revealed that the simulated growing season shifted by approximately a month compared to the observations; however, the peak simulated GPP was closer to the observations when calibrated crop parameters were used (Figure S19 in Supporting Information S1). The seasonal shift is likely due to the usage of GSWP3 meteorological forcing instead of the site specific forcing used for calibration results. The shift in the simulated growing season also occurred in monthly LE for the three calibration sites and three additional sites in the US-Midwest were LE is routinely measured (Figure S20 in Supporting Information S1).

Comparison of observed GPP with simulated annual GPP using calibrated but spatially invariant crop parameters (Set1, Set2, and Set3) reveals similar spatial patterns among the three regions. All three set of regional simulations overestimated GPP for most of the study region except for grid cells in Kentucky and Missouri (Figure S21 in Supporting Information S1). The absolute average difference in simulated and FluxCom based annual GPP from 2001 to 2010 was 29%, 22%, and 27% for Set1, Set2, and Set3, respectively. The absolute average difference for Set2 (22%) was slightly lower than for the composite set (25%). This is because Set2 simulates lower annual GPP for the entire US-Midwest region compared to the other Sets (Figures 6b–6d) bringing it closer to the Flux-Com estimates of annual GPP (Figure S22 in Supporting Information S1) that is lower than simulated estimates for most of the agriculturally intensive US-Midwest (Figure 7).

#### 4. Discussion

We found that spatially varying a small number of parameters had a large impact on carbon and energy fluxes; in particular, parameter optimization, and the use of spatially varying parameters, generally reduced the bias



# Journal of Geophysical Research: Biogeosciences

#### 10.1029/2022JG007187



**Figure 8.** Comparison of simulated and observed annual GPP at AmeriFlux calibration/validation sites: The simulated annual GPP was obtained from the regional run by identifying grid cells closest to the observation sites. Observed annual GPP are based on observations for both calibration and validation years and simulated GPP are based on average of 10 years of transient runs from 2001 to 2010. The default simulation utilized ELM default crop parameters while the composite simulation utilized parameters based on US-Ne3 calibration for the Northern Rockies, based on US-Ro1 calibration for the Upper Midwest, and based on US-UiC for the Ohio Valley (Figure 1).

between simulated and observed fluxes. These results have implications for optimal model parameterization, the importance of considering spatial variability in parameters as well as implementing crop rotation, and pathways to addressing known existing model limitations in the future.

#### 4.1. Model Parameterization: Optimization and Spatial Variability

The optimized parameter values estimated in this study differ across crop-sites but are within previously observed or modeled ranges. For instance, globally observational estimates of corn slatop have varied between 0.015 and 0.035 ( $m^2g^{-1}$ ) (Amanullah et al., 2007; Mohammadi, 2007; H. Zhou et al., 2020). In this study, the calibrated value of corn slatop range between 0.03 and 0.08 ( $m^2gC^{-1}$ ), that is equivalent to 0.014–0.034 ( $m^2g^{-1}$ ) (assuming the leaf carbon content is 45% of leaf weight) and falls within the observed range. Similarly, our calibrated value of hybgdd for soybean (1400), although significantly lower than the default of 1900, is similar to the values used for soybean in the US Midwest by Bilionis et al. (2015). Finally, the optimized value of mbbopt for corn is slightly higher than 4, except for corn at US-Ro1, that is consistent with the ELM default of 4. However, for soybean the optimized values of mbbopt is less than 7 (Table 5) that is lower than the ELM default of 9 for c3 plants. Similar to our findings, Duarte et al. (2017) found that lowering mbbopt from the c3 default of 9 to 6 better captured GPP and LE in coniferous forest in the northwestern US. In summary, the optimized parameter values identified here for slatop, hybgdd, and mbbopt improved flux estimation from the three calibration sites and are also similar in magnitude to values reported in prior observational or model studies.

One of the primary lessons from our analysis is that using constant crop parameters instead of spatially varying parameters can result in under or overestimation of regional fluxes, to the extent that it would be impossible to accurately (i.e., without significant spatial biases) capture the impact of crops on local and regional climate via biogeochemical and biophysical impacts on the land surface. Importantly, the impact of constant versus spatially varying parameters differs spatially and temporally (Figure S17 in Supporting Information S1) and cannot be estimated by simple scaling of the effects, as is true for a range of other systematic (as opposed to random) errors in ecosystem- to global-scale observations (Richardson et al., 2006) and models (T. Zhou et al., 2009). These systematic model errors are often the reason why ecophysiological and biogeochemical models have more difficulty reproducing spatial variability than overall means, at scales ranging from regional biomass and carbon

fluxes (Bond-Lamberty et al., 2007; Castanho et al., 2013) to global soil carbon pools (Todd-Brown et al., 2013). Here we use high quality observational data, from multiple AmeriFlux sites for managed croplands, to better capture the observed spatial variability in fluxes and quantify the parametric uncertainty introduced by using spatially-invariant parameters.

We found that parameter Set2 (based on data from US-Ro1) simulated annual GPP closest to FluxCom for the entire US-Midwest region (Figure S21 in Supporting Information S1), but this finding has two important caveats. First, satellite based estimates of cropland GPP, like FluxCom, have large uncertainty (Yuan et al., 2015), and therefore the lowest absolute average difference between FluxCom and a particular set does not mean that those optimized parameters are the best for the entire US-Midwest (Figure S22 in Supporting Information S1). Second, our results imply that the three AmeriFlux sites chosen for each region are not representative of the entire region, because sometimes fluxes estimated using optimized parameters from AmeriFlux site in another region are closer to observed fluxes; data from additional sites are thus needed to capture the spatial variability across the Midwest region.

#### 4.2. Importance of Climate and Crop Rotation

Non site-specific climatic forcing increased the difference between observed and simulated flux. Annual GPP observed at the three calibration sites was slightly different than simulated annual GPP in the regional run (Figure 8). This difference, despite calibration to the same data (Section 3.1), can be attributed to (a) GSWP3 forcing being used for the regional simulation compared to the site specific forcing utilized for calibration, and (b) GPP for all available years being used for estimating average annual GPP as opposed to only for the calibration years. These findings are generally consistent with previous work documenting the strong impact of climatic forcing data, at a regional scale, on carbon fluxes from forested region (Dorheim et al., 2022) and on above ground biomass estimation for mountainous region (Duarte et al., 2022). At a global scale, climate forcing can contribute more than half of total uncertainty in carbon cycle fluxes (Bonan et al., 2019) and can be the dominant driver of variability for the net ecosystem flux (Hardouin et al., 2022). The large contribution of climate forcing to total uncertainty also implies that the difference between observed and simulated fluxes can be further reduced by using a forcing data set that is better suited to the region.

The implementation of a realistic crop rotation capability in ELM allowed us to accurately capture the difference in peak flux magnitude between corn and soybean. Similar to our findings, Boas et al. (2021) found that realistic crop rotation improved LAI and latent heat flux estimation for field sites in Europe. Our findings suggest that simulating crop rotation, that is widely practiced in the Continental United States as well as globally (Sahajpal et al., 2014; Wallander, 2013), is important for accurately capturing feedback between human and agricultural ecosystem. Crop rotation representation will be even more critical in future integrated Earth System Models with enhanced human-climate feedback capabilities (Calvin & Bond-Lamberty, 2018; Thornton et al., 2017).

Although our analysis found that impact of crop rotation on annual GPP is negligible when scaled up to the gridcell, we argue that it is still important to represent crop rotation in ESMs as it affects biogeochemical cycles apart from the carbon fluxes; for example, it can result in markedly different yields from year to year (Figure 5). Similarly, crop rotation reduces fertilizer requirements and reduces nitrogen loss from agricultural systems.

#### 4.3. Model and Study Limitations

Both the ELM-Crop model and our study design have limitations that are important to note. In ELM, LAI is estimated as a product of specific leaf area parameter (slatop) and leaf carbon content. The underestimation of corn LAI for US-Ne3 and US-UiC in this study is likely caused by lower optimized value of corn slatop for these two sites (Table 6). Interestingly, prior studies using CLM have reported positive LAI bias which maybe due to higher slatop used in these studies compared to our study. For example, Peng et al. (2018) reported overestimation of corn and soybean LAI using CLM4 with an slatop value of 0.07. In our study, a higher calibrated value of slatop for corn at US-Ro1 also resulted in higher LAI, however, for this crop-site LAI observations are not available for comparison (Figure S10a in Supporting Information S1). We have more slatop data than almost any other trait (Kattge et al., 2020) but models are highly sensitive to it (Shiklomanov et al., 2020) and thus even small data/calibration problems in this area cascades throughout the model. Analogous to LAI, simulated

yield was also lower than the observed yield. Similar to our observations, lower corn yield was simulated using CLM4.5 (Peng et al., 2018) and CLM5.0 (D. L. Lombardozzi et al., 2020). Because of the close links between slatop, LAI, and photosynthesis, the start of the terrestrial chain of carbon processing, it is unsurprising that the underestimation of yield in this study and CLM are likely caused by the underestimation of above ground biomass (Peng et al., 2018).

Some of the limitations of the current study include performing site scale calibration and validation using limited QoIs. Future studies can enhance model performance by comparing against observations of above and below ground biomass; this is consistent with the argument of Keenan et al. (2012) who advocated for simultaneous calibration of models against diverse data streams. Another limitation of the current study is that for the regional simulations we did not account for how agricultural management practices of tillage, cover crops, crop residue management, and disease control can affect carbon and energy fluxes (Deryng et al., 2011; Dick et al., 1998). Estimating the impacts of these agricultural management practices on land fluxes and yield is beyond the scope of the current paper, but worth exploring in future studies.

Finally, and perhaps most fundamentally, ELM and most other global land models have plant and soil parameters that do not vary in space, time, or with forcing conditions such as light availability (Dohleman et al., 2009; Tian et al., 2015; Trócsányi et al., 2009; Van Esbroeck et al., 2003). Usage of constant parameters prohibits accurate estimation of various fluxes (T. Zhou et al., 2009) and crop yields (Osborne et al., 2015) and is a major limitation of the Earth System Models and a primary motivation for our analysis. This limitation can be addressed by modification of the model to read spatially variant crop parameters and generation of robust maps of parameters in space and time with well-defined errors. Finally, we need additional studies, similar to ours, that explore the magnitude of potential biases in agricultural ecosystems caused by spatially-invariant parameterizations in Earth System Models.

## 5. Conclusions

In this study, we implemented realistic agricultural management practice of crop rotation; calibrated and validated corn soybean rotation using multiple observations from the US-Midwest; and examined the impact of different parameterization schemes on carbon and energy fluxes.

We found that representation of agricultural management practice of crop rotation is important for studying the feedback between crops and climate and quantifying the impact of agriculture on energy fluxes and biogeochemical cycling. Our study shows that implementing crop rotation, and carefully calibrating crop parameters, improved estimation of site-level fluxes. We also found that the use of spatially variant crop parameters can have a large impact on carbon and energy fluxes. Such rigorous, spatially detailed approaches to crop modeling hold the potential to greatly improve flux estimation from agricultural regions.

Correctly representing the feedbacks between crops and climate is especially important for next generation ESMs that focus on improving the human-earth system interactions. Additionally, future studies focusing on calibrating corn and soybean for different regions or similar crops can optimize only the most sensitive parameters identified in this study for finding optimal parameter values. The reduced parameters can greatly reduce the surrogate models' dimensionality and improve their accuracy. It remains challenging to calibrate ESMs to multiple sites due to limited observational data availability, and our results emphasize the importance of observational networks such as FLUXNET (Baldocchi et al., 2001) and NEON (https://www.neonscience.org/) for ESMs.

## **Data Availability Statement**

Data from the AmeriFlux network US-Ne3 (Suyker, 2022), US-Ro1 (Baker & Griffis, 2018), US-UiC (Bernacchi, 2022), US-Bo1 (Meyers, 2016), US-Br1 (Prueger & Parkin, 2016), and US-IB1 (Matamala, 2019) were used in the creation of this manuscript. The E3SM model is described in detail at https://e3sm.org/. The source code for ELMv2 is archived and made publicly available at https://github.com/E3SM-Project/E3SM/ releases/tag/v2.0.0. All of the code supporting this paper is available at https://github.com/evasinha/Sinha-etal-2022-JGR-Bio and data supporting the paper is available at http://doi.org/10.5281/zenodo.7555458.



#### Acknowledgments

This research was supported as part of the Energy Exascale Earth System Model (E3SM) project, funded by the U.S. Department of Energy, Office of Science, Office of Biological, and Environmental Research. The Pacific Northwest National Laboratory is operated by Battelle for the US Department of Energy under Contract DE-AC05-76RLO1830. Dr. Katherine Calvin is currently detailed to the National Aeronautics and Space Administration. Dr. Calvin's contributions to this article occurred prior to her detail. The views expressed are her own and do not necessarily represent the views of the National Aeronautics and Space Administration or the United States Government. Funding for the AmeriFlux data portal was provided by the U.S. Department of Energy Office of Science. The field study conducted at the US-UiC site was funded by the DOE Center for Advanced Bioenergy and Bioproducts Innovation (U.S. Department of Energy, Office of Science, Office of Biological and Environmental Research under Award Number DE-SC0018420), the Energy Biosciences Institute at the University of Illinois Urbana-Champaign and the Global Change and Photosynthesis Research Unit of the United States Department of Agriculture/ Agricultural Research Service. We also thank two annonymous reviewers for their thoughtful comments that helped to significantly improve the manuscript.

#### References

- Amanullah, M. J. H., Nawab, K., & Ali, A. (2007). Response of specific leaf area (SLA), leaf area index (LAI) and leaf area ratio (LAR) of maize (zea mays L.) to plant density, rate and timing of nitrogen application. World Applied Sciences Journal, 2(3), 235–243.
- Baez-Gonzalez, A. D., Kiniry, J. R., Maas, S. J., Tiscareno, M. L., Macias C, J., Mendoza, J. L., et al. (2005). Large-area maize yield forecasting using leaf area index based yield model. Agronomy Journal, 97(2), 418–425. https://doi.org/10.2134/agronj2005.0418
- Baker, J., & Griffis, T. (2018). Ameriflux base US-Ro1 Rosemount- G21 [Dataset]. AMF. https://doi.org/10.17190/AMF/1246092
- Baldocchi, D., Falge, E., Gu, L., Olson, R., Hollinger, D., Running, S., et al. (2001). Fluxnet: A new tool to study the temporal and spatial variability of ecosystem-scale carbon dioxide, water vapor, and energy flux densities. *Bulletin of the American Meteorological Society*, 82(11), 2415–2434. https://doi.org/10.1175/1520-0477(2001)082(2415:FANTTS)2.3.CO;2

Bernacchi, C. (2022). AmeriFlux base US-UiC University of Illinois Maize-Soy [Dataset]. AMF. https://doi.org/10.17190/AMF/1846665

- Bilionis, I., Drewniak, B. A., & Constantinescu, E. M. (2015). Crop physiology calibration in the CLM. Geoscientific Model Development, 8(4), 1071–1083. https://doi.org/10.5194/gmd-8-1071-2015
- Boas, T., Bogena, H., Grünwald, T., Heinesch, B., Ryu, D., Schmidt, M., et al. (2021). Improving the representation of cropland sites in the community land model (CLM) version 5.0. Geoscientific Model Development, 14(1), 573–601. https://doi.org/10.5194/gmd-14-573-2021
- Bonan, G. B., Lombardozzi, D. L., Wieder, W. R., Oleson, K. W., Lawrence, D. M., Hoffman, F. M., & Collier, N. (2019). Model structure and climate data uncertainty in historical simulations of the terrestrial carbon cycle (1850–2014). *Global Biogeochemical Cycles*, 33(10), 1310–1326. https://doi.org/10.1029/2019GB006175
- Bond-Lamberty, B., Peckham, S. D., Ahl, D. E., & Gower, S. T. (2007). Fire as the dominant driver of central Canadian boreal forest carbon balance. *Nature*, 450(7166), 89–92. https://doi.org/10.1038/nature06272
- Bowles, T. M., Mooshammer, M., Socolar, Y., Calderón, F., Cavigelli, M. A., Culman, S. W., et al. (2020). Long-term evidence shows that crop-rotation diversification increases agricultural resilience to adverse growing conditions in North America. *One Earth*, 2(3), 284–293. https://doi.org/10.1016/j.oneear.2020.02.007
- Burrows, S., Maltrud, M., Yang, X., Zhu, Q., Jeffery, N., Shi, X., et al. (2020). The DOE E3SM v1. 1 biogeochemistry configuration: Description and simulated ecosystem-climate responses to historical changes in forcing. *Journal of Advances in Modeling Earth Systems*, 12(9), e2019MS001766. https://doi.org/10.1029/2019MS001766
- Calvin, K., & Bond-Lamberty, B. (2018). Integrated human-Earth system modeling—State of the science and future directions. *Environmental Research Letters*, 13(6), 063006. https://doi.org/10.1088/1748-9326/aac642
- Castanho, A. D. A., Coe, M. T., Costa, M. H., Malhi, Y., Galbraith, D., & Quesada, C. A. (2013). Improving simulated amazon forest biomass and productivity by including spatial variation in biophysical parameters. *Biogeosciences*, 10(4), 2255–2272. https://doi.org/10.5194/ bg-10-2255-2013
- Chen, M., Griffis, T. J., Baker, J. M., Wood, J. D., Meyers, T., & Suyker, A. (2018). Comparing crop growth and carbon budgets simulated across ameriflux agricultural sites using the community land model (CLM). Agricultural and Forest Meteorology, 256, 315–333. https://doi. org/10.1016/j.agrformet.2018.03.012
- Cheng, Y., Huang, M., Chen, M., Guan, K., Bernacchi, C., Peng, B., & Tan, Z. (2020). Parameterizing perennial bioenergy crops in version 5 of the community land model based on site-level observations in the central midwestern United States. *Journal of Advances in Modeling Earth* Systems, 12(1), e2019MS001719. https://doi.org/10.1029/2019MS001719
- Deryng, D., Sacks, W., Barford, C., & Ramankutty, N. (2011). Simulating the effects of climate and agricultural management practices on global crop yield. *Global Biogeochemical Cycles*, 25(2), GB2006. https://doi.org/10.1029/2009GB003765
- Dick, W., Blevins, R., Frye, W., Peters, S., Christenson, D., Pierce, F., & Vitosh, M. (1998). Impacts of agricultural management practices on c sequestration in forest-derived soils of the eastern Corn Belt. Soil and Tillage Research, 47(3–4), 235–244. https://doi.org/10.1016/ S0167-1987(98)00112-3
- Dohleman, F., Heaton, E., Leakey, A., & Long, S. (2009). Does greater leaf-level photosynthesis explain the larger solar energy conversion efficiency of miscanthus relative to switchgrass? *Plant, Cell and Environment, 32*(11), 1525–1537. https://doi.org/10.1111/j.1365-3040.2009.02017.x
- Dorheim, K., Gough, C. M., Haber, L. T., Mathes, K. C., Shiklomanov, A. N., & Bond-Lamberty, B. (2022). Climate drives modeled forest carbon cycling resistance and resilience in the upper great lakes region, USA. *Journal of Geophysical Research: Biogeosciences*, 127(1), e2021JG006587. https://doi.org/10.1029/2021JG006587
- Drewniak, B., Song, J., Prell, J., Kotamarthi, V., & Jacob, R. (2013). Modeling agriculture in the community land model. Geoscientific Model Development, 6(2), 495–515. https://doi.org/10.5194/gmd-6-495-2013
- Duarte, H. F., Raczka, B. M., Bowling, D. R., Wang, A., Buotte, P. C., & Lin, J. C. (2022). How can biosphere models simulate enough vegetation biomass in the mountains of the Western United States? Implications of meteorological forcing. *Environmental Modelling & Software*, 148, 105288. https://doi.org/10.1016/j.envsoft.2021.105288
- Duarte, H. F., Raczka, B. M., Ricciuto, D. M., Lin, J. C., Koven, C. D., Thornton, P. E., et al. (2017). Evaluating the community land model (CLM4. 5) at a coniferous forest site in northwestern United States using flux and carbon-isotope measurements. *Biogeosciences*, 14(18), 4315–4340. https://doi.org/10.5194/bg-14-4315-2017
- Golaz, J.-C., Van Roekel, L. P., Zheng, X., Roberts, A. F., Wolfe, J. D., Lin, W., et al. (2022). The DOE E3SM model version 2: Overview of the physical model and initial model evaluation. *Journal of Advances in Modeling Earth Systems*, 14(12), e2022MS003156. https://doi.org/10.1029/2022MS003156
- Hardouin, L., Delire, C., Decharme, B., Lawrence, D. M., Nabel, J. E., Brovkin, V., et al. (2022). Uncertainty in land carbon budget simulated by terrestrial biosphere models: The role of atmospheric forcing. *Environmental Research Letters*, 17(9), 094033. https://doi.org/10.1088/1748-9326/ac888d
- Hurtt, G. C., Chini, L., Sahajpal, R., Frolking, S., Bodirsky, B. L., Calvin, K., et al. (2020). Harmonization of global land use change and management for the period 850–2100 (LUH2) for CMIP6. *Geoscientific Model Development*, 13(11), 5425–5464. https://doi.org/10.5194/ gmd-13-5425-2020
- Iizumi, T., Tanaka, Y., Sakurai, G., Ishigooka, Y., & Yokozawa, M. (2014). Dependency of parameter values of a crop model on the spatial scale of simulation. Journal of Advances in Modeling Earth Systems, 6(3), 527–540. https://doi.org/10.1002/2014MS000311
- IPCC. (2019). Summary for policymakers. In P. R. Shukla, et al. (Eds.), Climate change and land: An IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems. Cambridge University Press. Retrieved from https://www.ipcc.ch/site/assets/uploads/sites/4/2020/02/SPM\_Updated-Jan20.pdf
- Jung, M., Schwalm, C., Migliavacca, M., Walther, S., Camps-Valls, G., Koirala, S., et al. (2020). Scaling carbon fluxes from eddy covariance sites to globe: Synthesis and evaluation of the fluxcom approach. *Biogeosciences*, 17(5), 1343–1365. https://doi.org/10.5194/bg-17-1343-2020



- Karlen, D. L., Hurley, E. G., Andrews, S. S., Cambardella, C. A., Meek, D. W., Duffy, M. D., & Mallarino, A. P. (2006). Crop rotation effects on soil quality at three northern corn/soybean belt locations. *Agronomy Journal*, 98(3), 484–495. https://doi.org/10.2134/agronj2005.0098 Kattge, J., Bönisch, G., Díaz, S., Lavorel, S., Prentice, I. C., Leadley, P., et al. (2020). Try plant trait database–enhanced coverage and open access.
- Global Change Biology, 26(1), 119–188. https://doi.org/10.1111/gcb.14904
  Keenan, T. F., Davidson, E., Moffat, A. M., Munger, W., & Richardson, A. D. (2012). Using model-data fusion to interpret past trends, and quantify uncertainties in future projections, of terrestrial ecosystem carbon cycling. *Global Change Biology*, 18(8), 2555–2569. https://doi. org/10.1111/j.1365-2486.2012.02684.x
- Lal, R., Delgado, J., Groffman, P., Millar, N., Dell, C., & Rotz, A. (2011). Management to mitigate and adapt to climate change. Journal of Soil and Water Conservation, 66(4), 276–285. https://doi.org/10.2489/jswc.66.4.276
- Lawrence, D. M., Fisher, R. A., Koven, C. D., Oleson, K. W., Swenson, S. C., Bonan, G., et al. (2019). The community land model version 5: Description of new features, benchmarking, and impact of forcing uncertainty. *Journal of Advances in Modeling Earth Systems*, 11(12), 4245–4287. https://doi.org/10.1029/2018MS001583
- Levis, S., Bonan, G. B., Kluzek, E., Thornton, P. E., Jones, A., Sacks, W. J., & Kucharik, C. J. (2012). Interactive crop management in the community Earth system model (CESM1): Seasonal influences on land–atmosphere fluxes. *Journal of Climate*, 25(14), 4839–4859. https:// doi.org/10.1175/JCLI-D-11-00446.1
- Li, Y., Yu, Z., Yang, S., Wang, G., Liu, X., Wang, C., et al. (2019). Impact of elevated CO<sub>2</sub> on C: N: P ratio among soybean cultivars. Science of the Total Environment, 694, 133784. https://doi.org/10.1016/j.scitotenv.2019.133784
- Liu, X., Chen, F., Barlage, M., Zhou, G., & Niyogi, D. (2016). Noah-MP-crop: Introducing dynamic crop growth in the Noah-MP land surface model. Journal of Geophysical Research: Atmospheres, 121(23), 13–953. https://doi.org/10.1002/2016JD025597
- Lobell, D., Bala, G., & Duffy, P. (2006). Biogeophysical impacts of cropland management changes on climate. *Geophysical Research Letters*, 33(6), L06708. https://doi.org/10.1029/2005GL025492
- Lombardozzi, D., Bonan, G., Wieder, W., Grandy, A., Morris, C., & Lawrence, D. (2018). Cover crops may cause winter warming in snow-covered regions. *Geophysical Research Letters*, 45(18), 9889–9897. https://doi.org/10.1029/2018GL079000
- Lombardozzi, D. L., Lu, Y., Lawrence, P. J., Lawrence, D. M., Swenson, S., Oleson, K. W., et al. (2020). Simulating agriculture in the community land model version 5. *Journal of Geophysical Research: Biogeosciences*, 125(8), e2019JG005529. https://doi.org/10.1029/2019JG005529
- Matamala, R. (2019). Ameriflux base US-IB1 Fermi National Accelerator Laboratory- Batavia (agricultural site) [Dataset]. AMF. https://doi. org/10.17190/AMF/1246065
- Mathison, C., Challinor, A. J., Deva, C., Falloon, P., Garrigues, S., Moulin, S., et al. (2021). Implementation of sequential cropping into JULESVN5. 2 land-surface model. *Geoscientific Model Development*, 14(1), 437–471. https://doi.org/10.5194/gmd-14-437-2021
- McDermid, S. S., Mearns, L. O., & Ruane, A. C. (2017). Representing agriculture in Earth System Models: Approaches and priorities for development. Journal of Advances in Modeling Earth Systems, 9(5), 2230–2265. https://doi.org/10.1002/2016MS000749
- Meyers, T. (2016). Ameriflux base US-Bo1 Bondville [Dataset]. AMF. https://doi.org/10.17190/AMF/1246036
- Mohammadi, G. R. (2007). Growth parameters enhancing the competitive ability of corn (Zea mays l.) against weeds. Weed Biology and Management, 7(4), 232–236. https://doi.org/10.1111/j.1445-6664.2007.00261.x
- Moore, C. E., Berardi, D. M., Blanc-Betes, E., Dracup, E. C., Egenriether, S., Gomez-Casanovas, N., et al. (2020). The carbon and nitrogen cycle impacts of reverting perennial bioenergy switchgrass to an annual maize crop rotation. GCB Bioenergy, 12(11), 941–954. https://doi. org/10.1111/gcbb.12743
- Moore, C. E., Gibson, C., Miao, G., Dracup, E., Gomez-Casanovas, N., Masters, M., et al. (2022). Substantial carbon loss respired from a corn-soybean agroecosystem highlights the importance of careful management as we adapt to changing climate. *Environmental Research Letters*, 17(5), 054029. https://doi.org/10.1088/1748-9326/ac661a
- Mueller, N. D., Butler, E. E., McKinnon, K. A., Rhines, A., Tingley, M., Holbrook, N. M., & Huybers, P. (2016). Cooling of us midwest summer temperature extremes from cropland intensification. *Nature Climate Change*, 6(3), 317–322. https://doi.org/10.1038/nclimate2825
- Nagasuga, K., Kadowaki, M., Uchida, S., Kaji, H., Fukunaga, A., & Umezaki, T. (2014). Effects of water condition on soybean (glycine max l.) plant growth after flowering. *Environment Control in Biology*, 52(4), 221–225. https://doi.org/10.2525/ecb.52.221
- NEON. (n.d.). (National ecological observatory network) NEON data portal. Retrieved from https://data.neonscience.org/home
- Nguy-Robertson, A., Gitelson, A., Peng, Y., Viña, A., Arkebauer, T., & Rundquist, D. (2012). Green leaf area index estimation in maize and soybean: Combining vegetation indices to achieve maximal sensitivity. *Agronomy Journal*, 104(5), 1336–1347. https://doi.org/10.2134/ agronj2012.0065
- Oleson, K., Lawrence, D., Bonan, G., Drewniak, B., Huang, M., Koven, C., et al. (2013). Technical description of version 4.5 of the community land model (CLM), NCAR technical note: Ncar/tn-503+ str. National Center for Atmospheric Research (NCAR). https://doi.org/10.5065/ D6RR1W7M
- Osborne, T., Gornall, J., Hooker, J., Williams, K., Wiltshire, A., Betts, R., & Wheeler, T. (2015). Jules-crop: A parametrisation of crops in the joint UK land environment simulator. *Geoscientific Model Development*, 8(4), 1139–1155. https://doi.org/10.5194/gmd-8-1139-2015
- Osborne, T., Slingo, J., Lawrence, D., & Wheeler, T. (2009). Examining the interaction of growing crops with local climate using a coupled crop-climate model. *Journal of Climate*, 22(6), 1393–1411. https://doi.org/10.1175/2008JCLI2494.1
- Pastorello, G., Trotta, C., Canfora, E., Chu, H., Christianson, D., Cheah, Y.-W., et al. (2020). The fluxnet2015 dataset and the oneflux processing pipeline for eddy covariance data. *Scientific Data*, 7(1), 1–27. https://doi.org/10.1038/s41597-020-0534-3
- Peng, B., Guan, K., Chen, M., Lawrence, D. M., Pokhrel, Y., Suyker, A., et al. (2018). Improving maize growth processes in the community land model: Implementation and evaluation. Agricultural and Forest Meteorology, 250–251, 64–89. https://doi.org/10.1016/j.agrformet.2017.11.012 Prueger, J., & Parkin, T. (2016). Ameriflux US-BR1 brooks field site 10-AMES [Dataset]. AMF. https://doi.org/10.17190/AMF/1246038
- Qian, Y., Wa, H., Yang, B., Golaz, J.-C., Harrop, B., Hou, Z., et al. (2018). Parametric sensitivity and uncertainty quantification in the version I of e3sm atmosphere model based on short perturbed parameter ensemble simulations. *Journal of Geophysical Research: Atmospheres*, 123(23), 13–046. https://doi.org/10.1029/2018JD028927
- Ricciuto, D. (2022). Offline land model testbed (OLMT). Retrieved from https://github.com/dmricciuto/OLMT
- Ricciuto, D., Sargsyan, K., & Thornton, P. (2018). The impact of parametric uncertainties on biogeochemistry in the E3SM land model. Journal of Advances in Modeling Earth Systems, 10(2), 297–319. https://doi.org/10.1002/2017MS000962
- Richardson, A. D., Hollinger, D. Y., Burba, G. G., Davis, K. J., Flanagan, L. B., Katul, G. G., et al. (2006). A multi-site analysis of random error in tower-based measurements of carbon and energy fluxes. Agricultural and Forest Meteorology, 136(1–2), 1–18. https://doi.org/10.1016/j. agrformet.2006.01.007
- Sahajpal, R., Zhang, X., Izaurralde, R. C., Gelfand, I., & Hurtt, G. C. (2014). Identifying representative crop rotation patterns and grassland loss in the US Western Corn Belt. Computers and Electronics in Agriculture, 108, 173–182. https://doi.org/10.1016/j.compag.2014.08.005

10.1029/2022JG007187

- Saltelli, A., Annoni, P., Azzini, I., Campolongo, F., Ratto, M., & Tarantola, S. (2010). Variance based sensitivity analysis of model output. Design and estimator for the total sensitivity index. *Computer Physics Communications*, 181(2), 259–270. https://doi.org/10.1016/j.cpc.2009.09.018
   Shiklomanov, A. N., Bond-Lamberty, B., Atkins, J. W., & Gough, C. M. (2020). Structure and parameter uncertainty in centennial projections of forest community structure and carbon cycling. Global Change Biology, 26(11), 6080–6096. https://doi.org/10.1111/gcb.15164
- Sinha, E., Calvin, K. V., Bond-Lamberty, B., Drewniak, B. A., Ricciuto, D. M., Sargsyan, K., et al. (2022). Modeling perennial bioenergy crops in the e3sm land model (ELMV2). Journal of Advances in Modeling Earth Systems, 15(1), e2022MS003171. https://doi. org/10.1029/2022MS003171
- Smith, R. G., Gross, K. L., & Robertson, G. P. (2008). Effects of crop diversity on agroecosystem function: Crop yield response. *Ecosystems*, 11(3), 355–366. https://doi.org/10.1007/s10021-008-9124-5
- Sobol, I. M. (2001). Global sensitivity indices for nonlinear mathematical models and their Monte Carlo estimates. *Mathematics and Computers in Simulation*, 55(1–3), 271–280. https://doi.org/10.1016/S0378-4754(00)00270-6
- Srivastava, A. C., Tiku, A. K., & Pal, M. (2006). Nitrogen and carbon partitioning in soybean under variable nitrogen supplies and acclimation to the prolonged action of elevated co2. Acta Physiologiae Plantarum, 28(2), 181–188. https://doi.org/10.1007/s11738-006-0045-7
- Stanger, T. F., & Lauer, J. G. (2008). Corn grain yield response to crop rotation and nitrogen over 35 years. *Agronomy Journal*, 100(3), 643–650. https://doi.org/10.2134/agronj2007.0280

Suyker, A. (2022). Ameriflux US-Ne3 Mead-rainfed maize-soybean rotation site [Dataset]. AMF. https://doi.org/10.17190/AMF/1246086 Thornton, P. E., Calvin, K., Jones, A. D., Di Vittorio, A. V., Bond-Lamberty, B., Chini, L., et al. (2017). Biospheric feedback effects in a synchronously coupled model of human and Earth systems. *Nature Climate Change*, 7(7), 496–500. https://doi.org/10.1038/nclimate3310

- Thornton, P. E., & Rosenbloom, N. A. (2005). Ecosystem model spin-up: Estimating steady state conditions in a coupled terrestrial carbon and nitrogen cycle model. *Ecological Modelling*, 189(1–2), 25–48. https://doi.org/10.1016/j.ecolmodel.2005.04.008
- Tian, S., Cacho, J. F., Youssef, M. A., Chescheir, G. M., & Nettles, J. E. (2015). Switchgrass growth and morphological changes under established pine-grass agroforestry systems in the lower coastal plain of North Carolina, United States. *Biomass and Bioenergy*, 83, 233–244. https://doi. org/10.1016/j.biombioe.2015.10.002
- Todd-Brown, K. E., Randerson, J. T., Post, W. M., Hoffman, F. M., Tarnocai, C., Schuur, E. A., & Allison, S. D. (2013). Causes of variation in soil carbon simulations from CMIP5 Earth system models and comparison with observations. *Biogeosciences*, 10(3), 1717–1736. https://doi. org/10.5194/bg-10-1717-2013
- Trócsányi, Z. K., Fieldsend, A., & Wolf, D. (2009). Yield and canopy characteristics of switchgrass (panicum virgatum l.) as influenced by cutting management. *Biomass and Bioenergy*, 33(3), 442–448. https://doi.org/10.1016/j.biombioe.2008.08.014
- Van Esbroeck, G., Hussey, M., & Sanderson, M. (2003). Variation between alamo and cave-in-rock switchgrass in response to photoperiod extension. *Crop Science*, 43(2), 639–643. https://doi.org/10.2135/cropsci2003.6390
- Venter, Z. S., Jacobs, K., & Hawkins, H.-J. (2016). The impact of crop rotation on soil microbial diversity: A meta-analysis. *Pedobiologia*, 59(4), 215–223. https://doi.org/10.1016/j.pedobi.2016.04.001
- Wallander, S. (2013). Crop rotation is far more common than continuous planting of major field crops. Retrieved from http://www.ers.usda.gov/ data-products/chart-gallery/gallery/chart-detail/?chartId=76619
- Wang, Q., Li, Y., & Alva, A., (2010). Cropping systems to improve carbon sequestration for mitigation of climate change. Journal of Environmental Protection, 1(03), 207–215. https://doi.org/10.4236/jep.2010.13025
- West, T. O., & Post, W. M. (2002). Soil organic carbon sequestration rates by tillage and crop rotation: A global data analysis. Soil Science Society of America Journal, 66(6), 1930–1946. https://doi.org/10.2136/sssaj2002.1930
- Wu, X., Vuichard, N., Ciais, P., Viovy, N., Noblet-Ducoudré, N. d., Wang, X., et al. (2016). Orchidee-crop (v0), a new process-based agro-land surface model: Model description and evaluation over Europe. *Geoscientific Model Development*, 9(2), 857–873. https://doi.org/10.5194/ gmd-9-857-2016
- Yuan, W., Cai, W., Nguy-Robertson, A. L., Fang, H., Suyker, A. E., Chen, Y., et al. (2015). Uncertainty in simulating gross primary production of cropland ecosystem from satellite-based models. Agricultural and Forest Meteorology, 207, 48–57. https://doi.org/10.1016/j. agrformet.2015.03.016
- Zhou, H., Zhou, G., He, Q., Zhou, L., Ji, Y., & Zhou, M. (2020). Environmental explanation of maize specific leaf area under varying water stress regimes. *Environmental and Experimental Botany*, 171, 103932. https://doi.org/10.1016/j.envexpbot.2019.103932
- Zhou, T., Shi, P., Hui, D., & Luo, Y. (2009). Global pattern of temperature sensitivity of soil heterotrophic respiration (Q<sub>10</sub>) and its implications for carbon-climate feedback. *Journal of Geophysical Research*, 114(G2), G02016. https://doi.org/10.1029/2008JG000850